Author's response to reviews

Title: Maternal Safety of the Delayed-Release Doxylamine and Pyridoxine Combination for Nausea and Vomiting of Pregnancy; A Randomized Placebo Controlled Trial

Authors:

Gideon Koren (pharmtox@sickkids.ca)
Shannon Clark (shclark@utmb.edu)
Gary D.V. Hankins (ghankins@utmb.edu)
Steven N. Caritis (scaritis@mail.magee.edu)
Jason G. Umans (jgu@georgetown.edu)
Menachem Miodovnik (menachem.miodovnik@medstar.net)
Donald R. Mattison (dm425@nih.gov)
Ilan Matok (ilanmatok@gmail.com)

Version: 8 Date: 2 February 2015

Author's response to reviews: see over
February 2, 2015

Dear Ian,

Thank you for your letter regarding our manuscript.

We have addressed all editorial requests as follows:

1) Authors suffixes – removed.
2) Abstract – follows the Journal’s style.
3) Trial registration has been added.
4) Competing interests have been added.
5) Authors’ contribution – have been added.
6) Acknowledgements – have been added.

Many thanks,

Gideon Koren, MD